Accepted Manuscript

Causes of child labour: Perceptions of rural and urban parents in Ghana

CHILDREN and YOUTH SERVICES REVIEW an international multidisciplinary proteor of the welfare of young proppie

Obed Adonteng-Kissi

PII: S0190-7409(18)30110-5

DOI: doi:10.1016/j.childyouth.2018.05.034

Reference: CYSR 3849

To appear in: Children and Youth Services Review

Received date: 17 February 2018
Revised date: 27 May 2018
Accepted date: 27 May 2018

Please cite this article as: Obed Adonteng-Kissi, Causes of child labour: Perceptions of rural and urban parents in Ghana. Cysr (2017), doi:10.1016/j.childyouth.2018.05.034

This is a PDF file of an unedited manuscript that has been accepted for publication. As a service to our customers we are providing this early version of the manuscript. The manuscript will undergo copyediting, typesetting, and review of the resulting proof before it is published in its final form. Please note that during the production process errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

Causes of Child Labour: Perceptions of Rural and Urban Parents in Ghana

Obed Adonteng-Kissi

PhD Student (Legal Studies), Department of Social Inquiry,
College of Arts, Social Sciences and Commerce
La Trobe University

VIC 3086 | Bundoora Campus, Australia

Phone: +61450720398

Email: dontosh@yahoo.co.uk

Abstract

The study focuses on parental perceptions of causes of child labour in rural and urban areas of Ghana. The research is grounded on qualitative research techniques by specifically utilising indepth interviews, focus groups, and participant observation to collect the necessary data for the study. In this approach, the data gathering happened in Phases 1 and 2 of the research study at the research sites in rural areas, and urban areas. The 60 participants included government officials, representatives from NGOs, and both parents whose children were involved in child labour and parents whose children were not involved. Much theoretical and empirical evidence is presented to argue that child labour has a multiplicity of causes including cultural practices, socialisation, poverty, and lack of mechanisation of work. This paper finds that the socio-cultural contexts of child labour play a critical role in children's involvement in farming in the rural area while poverty also contributes to children's engagement in artisanal fishing work in the slum urban communities in Ghana. This paper recommends that child labour must be tackled in a coordinated manner on a cross-sectoral basis and there is the need to adopt policies that would address the category of work that falls within worst forms of child labour (WFCL). Additionally, policy-makers and NGOs must consider the link between economic deprivation and child labour when implementing programmes designed to combat the problem.

List of Acronyms

AMA: Accra Metropolitan Assembly

BECE: Basic Education Certificate Examinations

CDW: Child Domestic Work

CHRAJ: Commission for Human Rights and Administrative Justice

CSO: Civil Society Organizations

DFID: Department for International Development

ESP: Education Strategic Plan

FCUBE: Free Compulsory Universal Basic Education

GDP: Gross Domestic Product GES: Ghana Education Service

GHC: New Ghana Cedi

GLSS: Ghana Living Standards Survey GPRS: Ghana Poverty Reduction Strategy

GSS: Ghana Statistical Service

IBRD: International Bank for Reconstruction and Development

ICESCR: International Covenant on Economic, Social and Cultural Rights

ICO: International Cocoa Organization

IITA: International Institute of Tropical Agriculture

ILO: International Labour Organization

JHS: Junior High School

MDG: Millennium Development Goals

MOE: Ministry of Education

NGO: Non-governmental organization

PTA: Parent Teacher Association

SDG: Sustainable Development Goal

SHS: Senior High School

TLM: Teaching and Learning Materials

TU: Tulane University

UDHR: The Universal Declaration of Human Rights UNDP: The United Nations Development Program

UNESCO: The United Nations Educational, Scientific and Cultural Organization

UNICEF: The United Nations Children and Educational Fund

USAID: The United States Agency for International Development

USLD: The United States Labor Department

WB: The World Bank

WFCL: Worst Forms of Child Labour

1. Introduction

Child labour is a legal concept with solid moral undercurrents rather than a dispassionate situation basically explained by statistics. Expressing legal standards in cross country data is The standards set by International Labour Organization (ILO) extremely challenging. Conventions express a global reference point (Diallo, Etienne & Mehran 2013). Nevertheless, the ILO statistics offer a fixed depiction of what appears to be a dynamic and extremely varied situation (Quattri & Watkins 2016). There is a distinction between child labour and child work in relation to domestic and global reporting perspectives. Child labour expresses a subcategory of employment or working situation (Quattri & Watkins 2016). The legal limits that distinguish child labour from "acceptable" child work is defined by three Conventions. The most rounded and extensively adopted global definition of the minimum age for admission to work is offered by ILO Convention 138 (1973). This Convention was approved by 169 countries and requires that ratifying countries set a minimum threshold of 15 years for admission to work, permitting a certain measure of discretion in developing countries to set a minimum threshold of 14 years. The age threshold is elevated to 18 years in relation to hazardous work. Hazardous work encompasses sectors such as mining and construction and other occupations such as metallic and electrical engineering activities, textile, wood-cutting, leather product machinery, and street vending. Convention 138 sets out to draw aspects of the distinction between child work and child labour. In participating in light work, the Convention sets the minimum threshold at 13 years but permits developing countries a certain measure of discretion to set the threshold at 12 years. Light work is explained as any harmless work by children that totals less than 14 hours a week. This is permissible work because it is not harmful to children's health and does not compromise their school attendance or other training activities.

The extent to which child labour is the result of economic forces or cultural norms and values is a major area of contention amongst scholars. In his study, "Getting Rid of Child Labour", Ahmed's empirical results evidently confirm that poverty is just an insignificant explanatory element behind the occurrence of child labour. In order of position, poverty came last among the seven causes of child labour. It is believed that child labour prevents children from attending school, hampers productive learning and refuses children a prospect to acquire the knowledge and skills they require to break free from poverty, and that their nations need to promote inclusive growth and human development (Quattri & Watkins 2016). However, there is rather fluidity in the interaction between child labour and school enrolment since some children combine work with education (Post 2018). There is a view that suggests that cultural attitudes towards childhood play a role in explaining the persistence of child labour (Zelizer 2005). Cultural values and practices contribute to the existence of child labour by shaping intra-family choices (Kabeer 1994) and mediating the effect of economic forces (Nieuwenhuys 2005). The focus of this paper is to provide a critical synthesis of the causes of child labour at the research sites in rural areas and urban areas. Examining parents' perceptions of the causes of child labour is an important topic of study, since parents plays a crucial role in the development of the child, and parents largely determine whether a given child will be engaged in work, and what kind of work they will perform (Bowlby 2008). The development of the child is grounded on interactions with others, and is moulded particularly by the child's parents.

The justification for comparison between rural and urban case study communities was underpinned by the fact that both areas offer interesting contexts for assessing the causes of

engaging children in economic activities in terms of parental knowledge and the effectiveness of the legal regime. The 1992 Constitution of Ghana introduced special provisions geared towards guaranteeing the rights of children as a unique group. This included an expansive set of rights designed to foster children's welfare and development. These constitutional provisions place obligations on Parliament to legislate to ensure and safeguard the welfare and development of children in Ghana. The legal regime on the protection of children was consolidated in 1998 in Ghana through the passage of the Children's Act (Act 560) 1998 (Twum-Danso 2009). The Act draws together previous child-related laws and makes necessary amendments which are tailored to meet UN and ILO benchmarks. On 5 December 2005, Ghana reached another major milestone in the protection of children's rights when legislation was passed to outlaw human trafficking within, to, from, and through Ghana ('Human Trafficking Act 694'). The Act's provisions were mainly motivated by the United Nations Protocol to prevent, suppress and punish trafficking in persons, particularly women and children (Palermo Protocol), which is the main international legal framework to fight trafficking (Sertich & Heemskerk 2011). The passage of this Act is also relevant to the protection of children from exploitative work. Despite the passage of the above new laws and policies to curb child labour in the two areas under study, parents still engage their children in farming and artisanal fishing work. Even though the rural and urban areas vary in size, population, nature of economic activities, infrastructural development and religious composition, they have a young age structure and a high incidence of child labour (Flora 2018; GSS 2014b). The demands on children to support family income in both rural and urban Ghana are such that a projected 21.8% of 5-17 year-olds are engaged in economic activities (GSS 2014b). Children involved in economic activities comprise 22.7% of boys and 20.8% of girls in rural and urban Ghana and the percentage of urban children in child labour was 12.4% compared to 30.2% of children in rural areas (GSS 2014b).

The lack of consensus in research on the relationship between child labour and its actual causes in the rural and urban areas may lead to questioning policy proposals that are intended to eliminate child labour. While the prevalence of child labour in the farming and artisanal fishing sectors have been extensively reported, little empirical consensus has been built on the causal conditions in households in developing countries (Adeoye et al. 2017). Some empirical results suggest that poverty is only a minor explanatory factor behind the incidence of child labour (Ahmed 1999). Therefore, Bhalotra (2003) observes that viewpoints that dwell on socialisation through parental perceptions of work or cultural practices, and efforts to shape economic models that relate to choices, deprivation and child labour have not been adequately researched. Furthermore, much research in rural and urban Ghana on causes of child labour tends to dwell on paid labour and underemphasise family work based on socialisation (Ray 2002). There is therefore a need to carry out a comparative analysis between rural and urban Ghana to investigate this assertion.

1.1 Why do parents put their children to work in rural and urban Ghana?

This research question is critical in directing the analysis of the study owing to the need to develop a scientific understanding of "causes of child labour" and formulate policies and programs to address them. The study reveals the following: Child labour has multiple, intersecting causes in both communities under investigation. In the rural areas, poverty, the socialisation of children, family livelihood and lack of mechanisation were the most common parental perceptions coming out of the data analysis, with socialisation acknowledged as the

most common cause of child labour. Mostly, all parental participants highlighted socialisation/culture as the main cause of child labour, while also pointing to other causes such as poverty and lack of mechanisation. In urban areas, the same category of causes of child labour were recognised by parental participants, with poverty being the foremost cause. Even though poverty is revealed as a perceived cause of child labour in both urban and rural settings, more parental participants (the focus group discussions, participant observation, stakeholder interviews and parental interviews) suggested poverty in the urban area than in the rural area.

The paper is presented in 7 sections, and sets out to explore parental perceptions of the causes of child labour in rural and urban areas of Ghana. In the first section, the aim and research question driving the paper are introduced. Section 2 discusses child labour in Ghana. Section 3 presents the concept of ecological perspectives, which provides a conceptual framework for the paper. Section 4 presents the methods and research design of the study, detailing what was done, how it was done, and why it was done and describes the study area. Section 5 presents the main findings that emerged from the study. Section 6 discusses the findings of the study. This paper is concluded in Section 7. Here, the research is assessed for what it has revealed about parental perceptions of the causes of child labour in rural and urban areas of Ghana.

2. Child Labour in Ghana

Child labour is an age-old practice which is widespread in all parts of Ghana (Asuming-Brempong et al. 2007). Children in Ghana are under pressure to make a contribution to family income, leading to 28% of children between the ages of 7 and 14 years engaging in remunerated economic activities (DFID 1998). They are required to work towards family subsistence at an early age in most regions. Child work forms part of the socialisation process of equipping children to take up adult skills in future life (ICO 2016; Löwe 2017). This socialisation process is in the form of an informal "apprenticeship", during which basic work ethic and skills are taught. For example, children accompany their parents to their farms to work as a means of passing on basic farming skills to them. They begin to gain insight into their parents' farming activities and become acquainted with the simple tasks of farming such as weeding, pod plucking, pod gathering and heaping, scooping of cocoa beans, carting of fermented beans and drying of beans (Ruf 2011). Children's farming activities also include carrying water for spraying, and carting of dry beans to sale hubs (Pasanen 2016). Tilling of the land and harvesting of crops which are harder work begin a little later when children are around 10 years of age (Löwe 2017).

A majority of children in rural areas are involved in economic activities in the agricultural sector, whilst children in urban areas primarily work in trade, child domestic work (CDW), fishing and construction. The number of children who were engaged in economic activities in the textile and footwear sectors is minimal compared to the number working as CDW (ILO, UNICEF & WB 2016). In Ghana, children below the age of 14 years constitute a significant part of the labour force in the cocoa and fishing industries (Kapoor 2017). Additionally, 40% of those working in rural areas are between 5-19 years of age (Mull & Kirkhorn 2005).

Ghana has a total child population of 2,236,124 in the cocoa-growing areas with an estimated 957,398 (42.8%) engaged in light work, 918,543 (41.1%) engaged in child labour and estimated 878,595 (39.3%) engaged in WFCL (Kapoor 2017). Furthermore, the percentage of children working in the agricultural sector declines with age, dropping to 57.1% in the 15–17 age

category, even though they are usually engaged more in WFCL (USLD/TU 2015). Additionally, the number of children in WFCL in cocoa production declined by 6% (from 0.93 million to 0.88 million) between 2008/09 and 2013/14 in Ghana (USLD/TU 2015). The WFCL in Ghana predominantly involve the carrying of loads, CDW, ritual servitude, commercial sexual exploitation, stone quarrying, small scale mining, fishing and cash crop agriculture (Asuming-Brempong et al. (2007).

The percentage of boys (83.7%) who were engaged in the agriculture, forestry and fishing industry was greater than the percentage of girls (70.0%) (GSS 2014a). In contrast, the percentage of girls who were involved in wholesale and retail trade (17.3%) was greater than the percentage of girls (7.9%). A slightly greater percentage of girls (4.4%) than boys (3.2%) were engaged in manufacturing. The percentage of girls in the accommodation and food service industry was 5.2% compared to 1.5% of boys in the same industry (GSS 2014a). It is estimated that there are 391,236 girls in cocoa-growing areas (or 36.8% of the total population of girls in this region) engaged in child labour, as opposed to 527,307 boys (or 63.2% of the total population of boys in cocoa-growing areas) (Kapoor 2017). Moreover, children are engaged in all adult tasks that are required of their gender at the age of 14 years (Hashim 2007).

Cocoa farming constitutes part of a larger agricultural livelihood strategy for households (Sumberg et al. 2012). The percentage of children who were involved in agriculture was greater in rural (87.5 %) than in urban (50.8%) areas (GSS 2014a). Greater percentages of children in rural savannah (92.7%) and rural forest (84.3%) were engaged in the agriculture sector compared to the rural coastal (74.2%) (GSS 2014a). Children in rural areas are more economically active than their counterparts in urban areas. Children in the rural areas are also more economically active at younger ages than other children in Ghana, with the highest proportion of children in the age 5-9 category (70%) engaged in some form of agricultural activity (Agbenyegah & Gockwoski 2002).

Approximately all the economically active children aged 5 to 14 years (99.8%) are engaged in some form of economic activity (GSS 2014a). The percentage of economically active children in rural areas is double the number in urban areas and this is true for males and females (GSS 2014a). The percentage of males aged 5 to 14 years who are economically active is higher than females in both urban and rural areas (GSS 2014a). Furthermore, the largest percentage (62.5%) of working children of both boys and girls aged 5–17 is to some extent involved in agricultural work (Mull & Kirkhorn 2005). The two principal harvest seasons are the small non-peak season from July to early September and the peak season from November to January. Ghanaian children (13%) between the ages of 5 and 14 are actively involved in farming activities (GSS 2007). There is intensive demand for labour on the Ghanaian cocoa farms which is seasonal. Ghanaian children (21.7%) are engaged in economic activities with a greater percentage (39.7%) located in the rural sector while 19.8% are in urban areas (Mull & Kirkhorn 2005).

3. Ecological Perspectives

The ecological perspective offers a conceptual framework for explaining people's way of life in the community context and further explains the community context itself (Trickett 2009). Furthermore, Trickett (2009) suggests that ecological perspective highlights how persons with

different cultural histories, skills, experiences, and resources react to personal difficulties, prospects, and constraints of the social contexts of significance to them. It points out that a historical perspective on the community context, highlighting the formative role of cultural and community history in explaining current community functioning (Trickett 2009). This approach is certainly reasonable since our knowledge and behaviours are guided by learned cultural scripts (Kral et al. 2011, p. 46). Consequently, knowledge about the local community is precondition and prelude to choices about what kinds of activities serve community goals and interests, and what individuals, groups, and social settings are most essential to the action goal according to the ecological perspective (Trickett 2009). Ecological perspective, therefore, evaluates how community traditions, resources, social structures and norms affect individual and group life. Therefore, in addressing a complicated socio-economic subject like child labour, it is extremely significant to understand and deal with human behaviour and problems in context. Many social interventions make minimal impact, or fail completely owing to the inability of policy-makers to understand the cultural and community contexts (Kral et al. 2011). This is so because the connoisseurs are found in the community. Interventions that are planned independent of culture are usually inclined to be ameliorative rather than transformative, and therefore tackle secondary causes rather than primary causes of a problem. Any intervention of a problem at an unsuitable level is probably going to disregard the most significant triggers of the problem (Jim 1992). Interventions that fail to understand culture usually have inadequate impact or fail because they do not change "the rules of the game". (Kral et al. 2011).

4. Methods

The study was designed utilising qualitative research techniques by specifically utilising in-depth interviews, focus groups, and participant observation to collect the necessary data for the study. Qualitative research has a specific contribution to make by assisting us produce valuable ideas about social issues at a series of levels, from that of individual perceptions through to that of how global systems work (Green & Thorogood 2018). Qualitative methods comprise a set of approaches for answering questions regarding what occurs, why and with what impacts at varied degrees. Purposive sampling method was utilised to assist in identifying the targets to answer the research questions (Mason 2017). Purposive sampling technique is a type of non-probability sampling that is most effective when one needs to study a certain cultural domain that involves well-informed experts (Lim, Shon & Yang 2018; Mason 2017). Furthermore, the investigator puts modalities in place to identify key informants who have the abilities and are willing to offer required information (Lucero et al. 2018; Mukherji & Albon 2018). The study specifically utilised focus groups, in-depth parent and stakeholder interviews, and participant observation techniques. The justification for applying these techniques is due to the lack of research in the study context and the need to obtain detail information to answer the research question. The detailed and complex data produced by semi-structure interviews is appropriate to offer clarifications (Draper & Swift 2011). A semi-structured interview schedule was applied to get information in the participants' own words, to provide a narration of situations and to produce

detailed information. My research question: "Why do parents put their children to work in rural and urban Ghana?" is supposed to offer varied accounts and consequently, language and content was made simple to comprehend by both professionals and parents alike. The interviews took place in varied settings comprising of the offices of the professionals and parents' own homes. Majority of the interview settings were chosen at each participant's convenience. The parental interview schedule comprised of 14 questions; stakeholders' interview schedule is made up of 20 questions; focus groups' interview schedule contained 13 questions; participants' observation interview schedule comprised of 12 questions and the interviews lasted between 30 and 90 minutes.

The sample size was 60 participants and they included government officials, representatives from NGOs, and both parents whose children were involved in child labour and parents whose children were not involved. I conducted a total of 10 semi-structured interviews with stakeholders from government agencies, departments and non-governmental organisations (5 stakeholders in each of the two areas under study). I moderated focus group discussions with 30 participants in total in the two areas under study. The focus groups were made up of parents, opinion leaders in the communities and traditional authorities. I recruited parents for the focus group discussions on parental attitudes towards child labour and human rights in the two communities under study. A total of 10 semi-structured interviews were conducted with parents whose children are involved in child labour activity and those whose children are not involved in this kind of work from each of the two communities under study. Finally, participant observations were also conducted to investigate children's natural working environment and to contextualise the empirical findings, helping to illuminate parental perceptions of human rights and child labour. A total of 10 semi-structured interviews were conducted with parents in the participant observations (5 parents in each of the two areas under study). The inclusion criterion for selecting participants for this study was being a parent who was a farm/boat/canoe owner with children engaged/or not engaged in child labour in the two communities under study or stakeholder in child welfare with expert knowledge.

I became the data collection instrument to assist in addressing the issue of validity and reliability (Bryman 2015). Therefore, my competence and interviewing skills and ability was the basis of reliability of the data gathered. Truthfully, my interview methods were improved and the quality of data improved as the research advanced (Ruckenstein 2014). Additionally, I improved interviewer reliability by conducting two pre-pilot interviews to get experience and the critical skills (Appleton 1995). Furthermore, I tackled the issue of reliability in the context of equipment used in the interview. I used a tape recorder to record all interviews, subsequently, improving reliability (Richards 2014). Therefore, I reported the findings in a way that was intended to express their tentative nature (Denscombe 2014). Likewise, I have evidently used auditability in this paper recounting each stage of the research procedure, amplifying and justifying what was done (Cope 2014).

I ran into possibilities of researchers bias to influence this study due to fact that I am a social scientist myself. Nevertheless, I was determined to overcome this by assuming a neutral stance and declining to give my own views during the data gathering process (Ormston et al.

2014). Hence, I checked and referred the data and the tentative interpretations to the research participants to approve of its credibility and accuracy at the end of the data analysis (Cope 2014). The goal was to establish if the findings recapitulated the participants' own experiences and explanations about the issue at stake. Similarly, the explanation of the analysis seems to be articulated in other areas. I triangulated my data by utilising different sources to confirm the developing findings (Carter et al. 2014). I considered my data and relevant literature which made me sure that the issue at stake as perceived by participants was accurately conveyed as possible. The data analysis was structured, comprising the creation of thematic matrices that systematically analysed data to ensure degrees of abstraction.

4.1 Limitations/Difficulties of the research

- The translation of participants' views from local language into English language that expresses participants' perceptions is one of the limitations of this study.
- Self-reporting of the participants brings about the problem of how much these views echoed the perceptions of the target population.
- I faced the problem of carrying out the study efficiently and effectively while upholding validity.
- Another problem was biased sample and biased returns because I applied purposive sampling in this qualitative research format.
- Furthermore, I did not give equal opportunity to many stakeholders in the target population to be interviewed.

4.2 Closing-the-Loop

- The application of propositions, for instance, was an avenue for us to reduce language of emotion and prove to the reader of my disconnection from the analysis.
- Furthermore, I standardised our interview schedule, the methods, findings and conclusion format which we designed to decrease rhetorical excess.
- Moreover, I used scientific language ostensibly to lessen multiple meanings from words in the interest of accuracy.
- Additionally, I conducted our study analysis to focus on the research language and we also managed the data in a relatively neutral fashion.

In this study, data gathering happened in Phases 1 and 2 of the research study at the research sites in rural areas (Ankaase, Anwiankwanta and Kensere), and urban areas (Jamestown, Korle Gonno and Chorkor). According to the 2010 *Population and Housing Census*, its total population is 20,451 (GSS 2012). Anwiankwanta, also known as Anweankwanta, is located at an elevation of 201 meters above sea level and has a total population of 29,748 (GSS 2012). Kensere is located at an elevation of 228 meters above sea level with a total population of 15,637 (GSS 2012). Jamestown, Korle Gonno and Chorkor are the three urban fishing communities under study in the Accra Metropolis. According to the 2010 *Population and Housing Census*, Jamestown has a population of 16,221 (GSS 2012). The second community, Korle Gonno is a well-established and typical indigenous coastal community known as Ga Mashie meaning King of Ga. According to the 2000 *Population and Housing Census*, it was estimated that the

population of Korle Gonno was 27,826 residents, projected to increase to 44,088 by 2008 (AMA 2008), while the total population of Chorkor stood at 23,853 (GSS 2002; Trickett 2009).

5. Results

My data indicate that a variety of factors contribute to the engagement of children in child labour in the areas under study. Key amongst them are the socialisation of children, poverty and lack of mechanisation of work. Although poverty is a significant factor in both rural and urban areas, interviews with parents also suggested that the relatively high prevalence of child labour in rural areas was because child work is a deeply rooted cultural practice, and because cultural structures in rural areas have been more sheltered from foreign intrusion than in urban areas.

Rural Areas

Socialisation Process

Many parents consider child work to be part of the socialisation process for children in local rural communities. It is part of the culture of many communities to socialise their children to take over from them in future. A parent said:

I believe I've to bring up my children in the way I was brought up. This is because my parents socialised, me to learn work ethics and that has made me who I am today. I don't see that when my children work in my farm during weekends and vacations, that can be called child labour.

Another parent explained:

My children help me to expand my farm. I also use the money I get from farming to look after them. It is also a way of training my children to take over from me when I am not around. I allow my children to work because of my children's future.

Many view the engagement of children in economic activities to enhance social competencies, build a sense of identity, strong emotional character, and gain career skills. The amount of time spent in a working context relates to better acquisition of the skills, knowledge and experiences linked with that context. The Director of CHRAJ explained:

Parents will engage their children in child labour because of the work socialisation process.

This is further supported by discussion in the focus groups, which suggested that work provides children with an avenue to develop their social and emotional competencies, including a strong work ethic. Some view children who are not working as uncultured and lazy. A parent said:

Whatever work I assign to my children is not intended to harm them but to train them to have strong work ethics that will in turn help them to have a better future. Children who go against cultural expectations by refusing to work are seen by society as deviants.

Participants in parental interviews noted a variety of benefits that children learn through work. As one parent explained:

Children use their abilities, express their interests, and exhibit creativity through work socialisation. Children also can work with people and to be useful to the society.

The more time they spend working on a task, the more skills, knowledge and experiences they gain. Parental participants in the participatory observation suggested that farmlands serve as a training ground to provide children with necessary skills that will be more useful to them in the future than attending school. As a Principal Labour Officer said:

Parents tell us that they know children in the community who have successfully completed their education, yet they don't have jobs. Some parents believe putting their children into farm work is worthwhile because their children develop their employment skills that get them ready jobs.

This view was supported by interviews with parents in rural areas. As one parent commented:

Why should I enrol my children in school if they're going to be unemployed after finishing? For me, I believe it will make more sense to engage them in my farming for them to acquire some skills that will help them in their future lives.

The issue of graduate unemployment is a hindrance to school enrolment since a significant number of parents mentioned that as the reason they put their children to farm work which will guarantee them a sustainable future income. Many parents believe that "good" children combine schooling and working. As one parent said:

I believe during the weekends children must make themselves available for farm work. Totally withdrawing from farming activities by a child is a sign of rebellion and stubbornness.

A similar view was expressed by the Director of CHRAJ:

In most parts of the rural areas it is within the culture for children to work to support their parents at an early age. If a child refuses to work, then society sees that child as a deviant, uncultured, disrespectful or even lazy.

This reflects the view that children have an obligation to their parents and to the community. Many focus group participants explained that the acquisition of abilities to work and solve problems raises children's capacity to produce goods and services. This investment of children's time in a work environment increases their lifetime incomes and promotes the growth of the local economy. As one parent said:

Children who don't support their parents on the farm are deviants. Parents won't toil and use their money on such children.

This development reinforces the economic and emotional ties in child work which has significant implications for parents' role in the payment of children's school fees. Child work is considered as a form of social support which is a major means by which parents and families get income to pay children's fees. They underpin the informal arrangements significant for a family's survival.

Poverty

Some participants in the study consider poverty to be a contributing factor to the engagement of children in work in Ankaase, Anwiankwanta and Kensere. Children need to work to bring in more money so that the household can meet the essential needs of all its members. Closely related to this, in farming communities, children sometimes need to work on the farm because the household cannot afford to pay for the services of farm labourers. A third possibility is that children need to work to pay for school fees, textbooks and transportation. The idea that there is a link between poverty and child labour was endorsed by a significant number of parents and stakeholders. A Director of CHRAJ said:

I think poverty plays a major role in child labour practice. No matter the number of intervention programmes that are brought in by NGOs and government, if parents are deprived, the practice will persist.

Many participants in the focus groups echoed this view. One parent explained:

The harsh economic situation demands that my children work to support me to take care of the entire family.

Another parent said:

We're all not on equal financial status. Some parents can care for the entire family without child work. Some of us too need our children to work to supplement family income.

Poverty has been a particular problem in recent years because of lower prices for cash crops which have put local farming families under pressure and led to a loss of local jobs and worsening deprivation. Many parents believe that engaging their children in child labour is the only way for families to avoid poverty. This point was reinforced by another parent who explained:

I don't know how I'm going to exit poverty in this difficult economic environment if my children do not support me on the farm to increase productivity. My children even appreciate the current predicament and they're happy to support me with farm work.

My field work also revealed that the financial strain on families in rural areas has a variety of further consequences. A parent said:

I can't put my children in school and access good health care if I don't allow my children to work to increase my income, so child work is necessary for me.

This clearly indicates that a further consequence of poverty is that some families cannot afford to enrol all their children in school, and instead prioritise the education of those deemed to be the most academically talented. As one parent told me:

I normally assess my children's level of intelligence to decide which of them to enrol in school. Children who fail their Basic Education Certificate Examination (BECE) which is their final examination join me on the farm to do farm work. I can't allow my children to sit at home without doing anything.

This implies that many parents assess their financial situation before deciding whether to put their children to work. This view was reinforced by interviews with stakeholders from the Labour Department. As a Registrar of CHRAJ explained:

Imagine a father who has a huge farm but has no money to hire the services of labourers. The farmer is likely to use his own children because farm work must go on anyway.

Parents also referred to the need for farm labour when explaining why they put their children to work:

Those of my children who aren't intelligent are put into farming activities to support those children who are in school. The farm work is huge, and I cannot do all by myself. We don't have the money to hire the services of labourers, so we're compelled to engage our children in work.

Some suggested that this need for labour was partly because many farmers could not afford to mechanise their farms. As a parent whom I interviewed in one of my participant observations commented:

The enormity and the tedious nature of our work demands many farm labourers. This is because I don't use machines here neither do I have the means of acquiring them. The only option left to me is to ask my own children to help me.

Focus group participants also explained that there is often a need for children to work in order to help finance their own education. A parent in a focus group explained:

Some children are eager to be in school, yet their parents can't afford it and therefore they want to work to get enough money to enrol. Let me tell you the truth. That's the only way such children can to enrol in school.

This view was echoed by some stakeholders. For example, a registrar of CHRAJ said:

When school authorities sack some of these kids for non-payment of school fees, some children feel they can work to support their education. Most often, the times that some children support their parents on the farm are the times that they are supposed to be in school.

Consistent with the idea that poverty is a key factor contributing to child labour, many parents pointed out that the farmers and households who are not involved in child labour are relatively well-off. As one focus group participant said:

Rich people do not engage the services of children because they have money to pay for the services of adult labourers. Hence, they hire the services of adults.

Commercial farms can also afford to make greater use of machinery, which reduces the need for labour. In contrast, there is a low level of mechanisation on family farms in Ghana. A relatively wealthy parent pointed out:

My children even don't know where all my farms are because they don't go to farm and I don't compel them to do so. It's therefore not true that all parents in

this community put their children to work. Many of the parents who do that, do that because of poverty.

Although NGOs in Ghana are increasingly taking up poverty alleviation strategies which were formerly carried out by state agencies, a number of participants noted that more needs to be done. A Director of CHRAJ said:

I think there are options open to us as a society. If the various NGOs working in these communities can support deprived parents financially, then children can be made to stop working and go back to school.

Overall, many the parents and stakeholders who were interviewed believed that poverty was a major factor in contributing to child labour in Ghana. However, the idea that poverty is the main cause of child labour was also challenged by some participants who argued that social relief programmes such as Free Compulsory Universal Basic Education (FCUBE), School Feeding Programme, Capitation Grant, Free School Uniforms and Exercise Books have made school affordable for most families. For example, a Principal Labour Officer said:

Mostly, child labour is blamed on poverty in these communities. However, I believe poverty is only used as an excuse. Parents have enough resources to enrol their children in school without engaging them in child labour.

State agencies offer incentives to parents who remove their children from child labour practices and enrol them in school. In view of this, a Labour Officer observed that some parents deliberately withdraw their children from school and put them into work so that they can then receive the incentive payment when they are subsequently re-enrolled. This assertion is reinforced by literature elsewhere (Skoufias et al. 2001).

Long Distance between Communities and Schools

The long distance between schools and residences in many rural communities also contributes to child labour because it is easier for children to work on a nearby farm than it is to attend school. A parent said:

I don't see why I should enrol my child in school if the school is far apart from my community. How does the government expect our children to get to school? Our farms are just around the community, so we can just as well train our children to be farmers.

This opinion was reinforced by other stakeholders. For example, a Labour Officer in the rural area said:

Distance makes it difficult for children to go to school regularly. What is now being done is to build community schools for children to enable them to be in school.

Participants suggested that if schools are built in every community, parents will not have the excuse to withdraw their children from school and engage them in child labour.

Urban Areas

Poverty

According to a Director of the Social Welfare Department, child labour in urban areas is a poverty issue. The Director explained that most of the children who go to sea and sell on the streets come from deprived homes. Children must assist their families by working to supplement their families' income. A Director of the Child Resource Institute expressed a similar view:

Basically, the problem is about poverty. Many livelihoods of parents are hand to mouth businesses, so they are unable to pay for their school fees. That is why you see many children at home and not going to school. We are working with limited facilities, so we are unable to enrol as many children as we should in our school.

Many parents interviewed in urban areas suggested that parents engage their children in work because of deprivation. A parent said:

I don't get enough income from my fishing business to enable me to shoulder all the family responsibilities, so my children need to help me to increase our family income.

However, other participants believed that although many parents would want to enrol their children in school, they are not prepared to make the economic sacrifices required to achieve this. Some also have misplaced priorities when it comes to the allocation of their financial resources. As one parent said during an interview:

Some parents spend their income on parties and funeral clothes rather than their children's education. For me it's a matter of spending on things that one believes really matters.

This view was to some extent strengthened by a Deputy Director of the Social Welfare Department who said:

Some parents often opt for the option of least resistance, that is, putting their children into work instead of school. Given the choice where all parents have the capacity to enrol their children in school, all parents will want their children to be in school. Children working is not a preference for parents.

Others point out that among poor families' children enrolment in school is often dependent on them being able to go to sea with their parents. One focus group participant said:

Children go to sea and sell for their parents. Let's be clear and honest on some of these issues. Parents push many of their children you see going to sea or selling in this community into work. We see children who are five, six, seven, eight, years selling around. Their parents send them, and they don't sell for themselves. Children sell because of poverty; parents need money to put their children in school that is why they put them into work.

Education in Ghana is supposed to be free as provided under the Free Compulsory Universal Basic Education (FCUBE) policy but, it is not free because parents are required to pay different kinds of school levies. The various levies that schools charge parents have watered down the

substance of FCUBE, along with other welfare programmes that were supposed to make education more accessible, such as the Capitation Grant and the School Feeding Programme. This view is shared by the Deputy Director of Social Welfare, who drew on his own experience to illustrate the point:

I have a house help who's enrolled in a public school. I pay various fees not less than GHC 50 every term. This amount of money cannot be borne by the average Ghanaian. The school feeding programme on the other hand is limited to some few schools since it is on a pilot basis.

Parents who cannot afford to pay the various school levies are likely to engage their children in economic activities to help them meet this cost in the future. However, there is no guarantee that they will earn enough to be able to afford to put their children back to school. A parent explained:

Education is supposed to be free, yet parents are asked to pay development levy, parent teacher association levy and library levy, so some parents engage their children in work to enable them pay for all these levies.

As in the rural areas, there were some urban participants who believed that poverty is no excuse for failing to enrol children in school. Parents should find the resources for this, even if it would mean selling family assets. As one parent commented:

I sold my parcel of land to pay for my children's Senior High School education when he qualified from the Junior High School to enter SHS (i.e. Senior High School).

However, once again, the dominant view among participants was that poverty was a key factor in explaining why children are engaged in child labour.

Cheap Labour

The need for cheap labour is another reason for engaging children in fishing. Children have little or no bargaining skills when it comes to employment matters. Some adults therefore take advantage of children's vulnerability in negotiations to make more profits. A focus group participant said:

Some parents and fishermen consider children as cheap labour. Many parents want to make more profits.

Adults charge higher rates for their services, so canoe/boat owners will often prefer to hire children instead. Often, they will not want to give a percentage of their catch to the child labourers on board either, which is the practice when adults are employed. As one parent said:

Boys of 8 to 9 years will not demand much from fishermen when there is a catch. Most children are ready to accept a meagre amount of money as remuneration.

Some stakeholders explained that adults who engage children in child labour activities are just lazy. A Principal Labour Officer said:

Some adults do not want to work hard. Other adults also use children to work for them because their labour is very cheap.

Many participants believe that the total elimination of children from fishing will affect the local economy. A focus group participant suggested:

Our family businesses will collapse if we take our children out of fishing. It is part of our culture for children to get deeply involved in what we (parents) do.

Some adults can intimidate children in bargaining for their services. However, those adults lose sight of the fact that a well-developed adult can contribute more than a child. It is also common for children in Ghana to be raised by family members or other adults who are not their biological parents, and many participants suggested that children in these situations were particularly vulnerable. Thus, many children living with relatives other than their biological parents are especially likely to be involved in child labour and suffer other forms of abuse. As one said:

The kind of treatment some children go through in their homes portray the insensibilities of their guardians. It is also an indication that such children aren't living with their biological parents.

After returning home from fishing, many are then expected to perform difficult household tasks before going to bed. This was reinforced by a participant in focus group who said:

It's known in this community that most of the children who go through various forms of abuses don't live with their biological parents

In Ghana, foster care is under-regulated and this leaves children whose biological parents are not present particularly vulnerable to child labour (Kuyini et al. 2009).

Lack of Mechanisation

Lack of mechanisation is another factor contributing to child labour in urban areas. The artisanal nature of fishing makes the work difficult to do. Parents therefore need more assistance to be able to carry out their trade. As one parent explained:

Our work (fishing) is not mechanised so it is quite difficult and therefore, we need the support of our children to be able to do our work. I have five children, I have put the two most brilliant ones in school, and the other three are working to support me look after the brilliant ones. I can't also bring in other people's children to work with me. It's better I work with my own children.

Focus group participants observed that child labour is widespread in the area under study although the government has outlawed it. A focus group participant said:

This is because there is inadequate mechanisation of artisanal fishing industry. The work is quite difficult, and fishermen need more hands.

Other stakeholders also believed that a lack of mechanisation contributes to child labour. A headmaster of a school said:

I think many parents are using their children to work because their work is not mechanised. The State needs to support parents to mechanise their work to help eliminate child labour.

Participants suggested that if government can help artisanal fishermen to mechanise their activities, child labour may be reduced.

Socialisation

Parents in the participant observation suggested that fishing activities serve as a vehicle to adequately prepare children for the future. Thus, children are engaged in work as part of the cultural socialisation process. Parents see child work as a training ground for children to understudy their parents in the work environment, preparing them to carry on the family business in the future. A parental participant in one of the participatory observation groups said:

I don't believe when my child work with me during his leisure times in my fishing business it is child labour. Engaging my children in fishing is simply a traditional system of schooling which doesn't violate the rights of my children.

According to stakeholders, the involvement of children in fishing is part of the culture of the communities and an important means for them to develop psychosocial skills. Involvement in child-friendly work helps children learn to take turns, be helpful, share, and care for others. A parent said:

I can boldly tell you that the practice is influenced by the culture of the people. In the olden days owners of canoes always wanted to have male children to take them to sea. If fishermen bring forth only female children that means anybody who marries their daughters will inherit their canoes. Society consider these male children as helpers of the family business. I didn't go to school because the culture had some form of influence on my education. I was always interested in working.

Some of the household work enables children to understudy their parents to become responsible adults capable of managing themselves. A parental participant in one of the participant observations said:

I want my biological children to benefit from my business. I take my children to sea as part of the succession plan to take over from me.

Some stakeholders also observed that some children who participate in fishing are engaging in both work and recreation. While earning an income to help support their household, they also can play and swim for leisure. According to a Deputy Director of the Social Welfare Department:

The recreational activities of the people should enable us to understand that child labour is not necessarily about poverty.

This is another reason why an understanding of the social/cultural context is important in understanding child labour. However, many participants also believed that some children are

made to perform an excessive amount of work that goes beyond what they need for the purpose of socialisation. A focus group participant explained:

I think some parents overburden children with work because our society see child work as part of the socialisation process.

This assertion is reinforced by many more participants in the parental interviews. For example, a parent explained:

Sometimes it's difficult to confront or report a parent to the police if you see a child being overburden with work in this community. This is because we all understand the socialisation principle is a core value in our society.

This indicates that although most parents believed that sending children to work was an important part of the socialisation process, some also believed that this practice was abused by some adults who made children perform exploitative and dangerous forms of work.

Illiteracy and Ignorance

Lack of parental education is another factor that contributes to child labour. A Director of the Social Welfare Department explained:

If a child has an educated parent, no matter the economic hardship, they will still find a way of enrolling their children in school. There are public schools, which do not cost so high as compared to the private schools. Parents can therefore enrol their children in the public schools where there are safety nets like the FCUBE, Capitation grants, school feeding programme, free school uniforms, free exercise books etc.

Many focus group participants also agreed that the ignorance and irresponsibility of some parents contributes to the problem of child labour. A parent reiterated this point:

For me, I think ignorance brings about irresponsibility. How can you tell me that parents who use their money to do unnecessary things can't pay for their children's schools fees? They simply don't know the importance of schooling, so they abandon their children and put their money in wrong things.

Many participants in the focus groups noted that some parents abandon their children because of strained relations with their partner. Other parents have enough savings to buy clothes for ceremonies, yet they are unable to pay school fees for their children.

6. Discussions

The main aim of this paper is to explain the reasons children become involved in child labour in rural and urban areas of Ghana. Child labour has multiple, intersecting causes in both communities under investigation. Although some economists and sociologists believe that child labour is the result of economic deprivation, this paper find that the socio-cultural contexts of child labour also play a critical role in children's involvement in fishing and farming work in Ghana. Much empirical evidence is presented to demonstrate that child labour has a multiplicity

of causes including cultural practices, socialisation, poverty, and lack of mechanisation of work (Ersado 2005). The economic structure in rural Ghana is different from urban areas. The rural area under study is agrarian local economy. Many of the local population grow mostly cocoa, cassava, plantain, vegetables, fruits and other staples. Household survival depend mostly on agrarian activities consisting of subsistence and cocoa which is a major cash crop in the communities under investigation. These crops feed the local households and provide reasonable amount of money for them. In the urban areas, the local population are engaged in artisanal fishing which is associated with lower incomes and income instability owing to extensive seasonal fluctuations in fish catch. Akaba and Akuamoah-Boateng (2018) echoes this claim by pointing to fluctuations in proceeds in artisanal fishing. Artisanal fishermen in these urban communities are largely poor since they have smaller boats and gear, and catch relatively smaller quantities of fish compared to the large-scale commercial fishing vessels (Grainger & Parker 2013). The fishing vessel often utilised by artisanal fishermen is a big dug-out wooden canoe. Artisanal fishing communities have higher rates of fertility and population growth rates (GSS 2015b). Higher fertility in artisanal fishing communities has been attributed to the high demands for labour in fishing and the role of kin-based labour (Marquette et al. 2002).

I differentiate between the rural communities where the socio-cultural consideration applies and the urban communities under study where poverty is the major cause of child labour. In the rural areas, the socialisation of children, poverty, family livelihood and lack of mechanisation were the most common explanations emerging from the data analysis, with socialisation identified as the most common reason for child labour. Engaging children in child labour to socialise them is not limited to Ghana but prevalent in other developing countries. In other developing countries in which farming and artisanal fishing are main industries and school education is not supplied sufficiently, as found in this study, making children succeed the parents' job and, for this reason, teaching children the job by parents may be the best way for their subsistence. In the rural communities under study, for example, if household needs GHC 100 to survive and school fees is GHC 20 and the parents earn GHC 120, children do not have to work for the family's survival and therefore, they can attend school. However, children may work and not go to school even in this situation. It can be assumed that the socialisation by the parents has occurred and that school education is lost due to the socialisation process in this instance. In the rural communities under study, socialisation is not a completely hopeless situation since children do not become impoverished when they become adults. Nevertheless, it becomes difficult for the economic situation of generations of such a family to be improved. Children cannot accumulate human capital that can be used in the more productive industry without school education. Therefore, they cannot save towards the future and as a result they cannot provide their children with adequate education. The country will find it difficult to exit poverty if it is comprised of such households.

The high level and continuous engagement in child labour presents a difficulty in the achievement of the Sustainable Development Goals (SDGs) set for 2030 (Quattri & Watkins 2016). These goals include the eradication of poverty, decent quality learning for all children up to secondary school level, reduced inequality and the creation of decent jobs (SDG 2017). High level children engagement in child labour is in conflict with these targets, particularly those connected with education (Guarcello, Lyon & Valdivia 2015). There is long-term negative impact of utilising child labour as a socialisation process in Ghana and other parts of the developing world. Besides the exposure to risk of injury, these children are deprived of the opportunity to acquire what Amartya Sen describes as 'human capabilities' — the knowledge,

skills and competencies needed to expand choices and extend opportunity (Sen 1997). Children who make early entry into the world of uncertain, untrained, low-paid work are improbably going to accrue the education they require to get decent work and exit poverty across generations. Widespread child labour is a barrier to human development since it undermines the human capital on which dynamic and inclusive economic growth, increasing productivity and social progress is based. It is no accident that child labour is firmly connected with low income both across and within countries.

In urban areas, the same causes of child labour were identified, with poverty being the leading cause. Although poverty is mentioned as a perceived cause of child labour in both urban and rural settings, more participants (the focus group discussions, participant observation, stakeholder interviews and parental interviews) indicated poverty in the urban area compared with the rural area. In the urban area for example, if the family needs GHC 100 to survive but parents earn only GHC 70, children must work to earn at least additional GHC 30. Therefore, child labour in the urban communities under study is because of poverty and cannot be taken as a socialisation process even if children have an opportunity to learn how to engage in their parents' work while engaging their children in child labour. Quattri and Watkins (2016) points out that children are pushed into child labour by economic deprivation reinforcing the view that parents make the decision to engage their children in child labour as a 'distress choice' meant to meet a minimum income level. Moreover, even if education is free, children will not receive education because they must work and cannot attend school. In this context, children forfeit their schooling owing to poverty and not due to their socialisation process.

The results of my analysis suggest that more children in the urban areas under investigation work because of poverty than the rural areas under study. Additionally, more parents in the urban area under study view poverty as the major cause of child labour than the rural area under investigation. Although, poverty as a reason for child labour holds in both urban and rural areas, but it is more pronounced in urban area. In contrast, the 2016 Ghana Poverty and Inequality Report, suggests that households in urban areas continue to have a lower average rate of poverty than those in rural areas (Cooke, Hague & McKay 2016). The report further indicate that urban poverty has declined in present times more rapidly than rural poverty leading to a doubling of the gap between urban and rural areas. Presently, rural poverty is nearly 4 times as higher than urban poverty compared to 1990s when rural poverty used to be two times higher than urban poverty (Cooke, Hague & McKay 2016). There seems to be tension between my findings and the 2016 Ghana Poverty and Inequality Report. This divergent interpretation from my study is based on participant observation, focus group discussions, and stakeholder and parental interviews. In seeking explanations for these divergent results, I uncovered in my study that contextual factors need to be taken into consideration to be able to explain the contradiction. I explain the elements underpinning this trend and demonstrate that it is the extent to which geographic adjustment of economic welfare in the Ghana influences the measurement of poverty. Additionally, Nolan, Waldfogel and Wimer (2017) find that there is a more communal way of living in the rural areas than urban areas which serves as robust antipoverty system and safety net. This is the case in both geographically unadjusted and adjusted measures of poverty. Furthermore, rural communities below 200 percent of the poverty line are more likely to receive support through their communal way of living (Nolan, Waldfogel & Wimer 2017). The communal way of living is more effective antipoverty and Social Security system in the rural areas than their urban counterparts, and that this is the situation across racial/ethnic groups (Nolan, Waldfogel & Wimer 2017). This finding is consistent with literature elsewhere in the

world. For example, Ye (2017) explains that urban communities are heterogeneous "melting pots" drawing people from diverse ethnic, cultural and linguistic backgrounds. Migration has brought new diversities to enrich as well as complicate the dynamics of urban communities. This has also resulted in a renewed interest in the possibility of the regular intercultural encounter in shared places. There is also a higher percentage of female-headed families in urban communities, which are more vulnerable to poverty. The inability of deprived households to cater for the family motivates them to engage their children in economic activities. Enrolling children in school is a luxury that deprived families are often unable to afford.

Furthermore, participants in both the urban and rural areas engage their children in work for reasons of socialisation. This is consistent with the findings of a recent study by Adu-Gyamfi (2014) who found solid evidence that children in Ghana engage in work for cultural and socialisation reasons. Child labour may be considered harmful to people from Western cultures, but traditionally *child labour* has played a crucial role in the rearing of Ghanaian children, helping them learn the importance of work, alongside practical skills that are regarded as essential to being a valued member of the family and community. Childhood studies in anthropology have claimed that "childhood is socially constructed and it changes depending on the historical and cultural setting" (Montgomery 2008, p. 51). The findings of the study support this view and show that a proper understanding of the causes of child labour must pay attention to cultural variations in how the idea of "childhood" is constructed.

As the paper has established, poverty is the factor most likely to be identified by parents as a cause of child labour in the artisanal fishing slums in the urban area. This means that in addressing the problem, the government should be careful not to implement policies that widen inequality and leave poorer families and children worse off. In many cultures in the developing world, work is viewed as a means of helping to socialise children, contributing to their personal development. However, according to parental participants in my study, there are limits to the kind of work children should be expected to perform. "Child labour" problems should be understood in the context of WFCL, hazardous work or the abuse of children in work, rather than work socialisation (White 1999).

7. Conclusion

In Ghana, child labour is a complex practice as in other developing countries, that must be tackled in a coordinated manner on a cross-sectoral basis. Any constructive effort will need combined methods that range from the control of labour markets, education, child welfare and varied strategies for poverty reduction. There is also the need for a framework of dialoguing which is intended to change attitudes. Based on the findings from the focus group discussions, participant observation, stakeholder interviews and parental interviews, an adequate policy response that includes: designing an enhanced evidence base approach to reflect the level of child labour and the schooling status of child labourers, concentrating particularly on urban fishing slums and conducting studies that reflect the reported experiences of children. Furthermore, there is the need to make the elimination of child labour a leading goal of education policy. In 2015, the overall expenditure on education to the Gross Domestic Product (GDP) was 3.9% (GSS 2015a). Therefore, there is the need to increase the total expenditure of education to about 4–5% of GDP, with a stronger focus on overall equity and education delivery in artisanal fishing slums of the urban areas.

This paper recommends improvement in the quality of education in artisanal fishing slum communities and promoting the school readiness of fishing slum-dwelling children through universal early childhood provision in fishing slums (Quattri & Watkins 2016). Further, there is the need to invest in the human resources required to make sure that regulatory agencies are equipped to carryout effective inspections. Additionally, punitive fines need to be imposed on employers found to be engaging under-age children in work. Again, inspection arrangements need to be reviewed for the artisanal fishing sector to ensure that fishermen comply with national laws.

It is also important that community schools for children be built in every locality to enable children to be in school to address the problem of schools being far from communities. This is because parents and stakeholders identified travel time as a factor discouraging children from regularly attending school. Issues relating to access, quality and teacher training also need to be addressed, and interviews again suggested that teacher quality discouraged parents from sending their children to school.

This paper recommends educational programmes for parents on the need to enrol their children in school and other vocational apprenticeships. This is intended to correct the perception that working is more beneficial for children than education and training. In some of the areas under study, as discussed in section 5, parents already had a good understanding of why WFCL is wrong, partly because of existing media campaigns. There is a need for various children's rights NGOs and government agencies responsible for children's welfare to collaborate on providing better education for parents, especially in the fishing and cocoa growing/farming communities (Figiel, Kuberska & Kufel 2014). Apart from government efforts to address WFCL, there is also the need for the design and implementation of scholarship schemes for academically talented children from disadvantaged families in these communities.

There is the need for steps to be taken to combat the conditions of poverty and economic vulnerability that contribute to the problem of child labour. Poverty itself should be officially declared as an infraction of the rights of the child nationwide. The Universal Declaration of Human Rights (UDHR) makes provision for the freedom from want. Additionally, Article 27 of the UNCRC acknowledges the right to "a standard of living adequate for the child's physical, mental, spiritual, moral and social development". This standard is acknowledged by the economic and cultural rights since 1966 in the International Covenant on Economic, Social and Cultural Rights (ICESCR). This reinforces the argument that poverty violates the rights of children by thwarting their optimal development. There is need for primary prevention interventions that focus on community poverty in general. The government should institute income transfers and/or subsidies for deprived households with children in school. This is critical for child labour reduction. The implementation of social relief programmes, such as free compulsory universal basic education (FCUBE), the school feeding programme, the capitation Grant, free school uniforms and free exercise books, should be strengthened to cover all children in Ghana. As discussed in section 5, education in Ghana is supposed to be free as provided under the Free Compulsory Universal Basic Education (FCUBE) policy but, it is not free because parents are required to pay different kinds of school levies. The various levies that schools charge parents have watered down the substance of FCUBE, along with other welfare programmes that were supposed to make education more accessible, such as the Capitation Grant and the School Feeding Programme. This would help take deprived children from work and enrol them in school. There is also the need for systemic and intervention programmes with legal

backing by the government of Ghana to offer professional social work programmes and services that go beyond the constitutional provision outlined in their national constitutions (Kaltenborn et al. 2017; Tsikata 2010).

This paper recommends the adoption of policies that would address the category of work that falls within WFCL. Additionally, policy-makers and NGOs must consider the link between economic deprivation and child labour when implementing programmes designed to combat the problem. Adopting more advanced forms of technology in agriculture and fishing would also help combat child labour. This was one of the factors that helped to curb the use of child labour in 19th century Europe. In Ghana today, mechanisation would mean that routine and repetitive tasks could be carried out by machines instead of children. This would increase the motivation for children to stay in school to improve their skills. The government could also provide local farmers and fishermen with credit facilities with flexible repayment terms to help them mechanise their operations or adopt labour-saving technologies (Cofie 2015; Yirga 2012).

References

- Adeoye, S, Agbonlahor, M, Ashaolu, O & Ugalahi, U 2017, 'Analysis of Child Labour Dimensions and Causes in Rural Farm Households of Ogun State, Nigeria', *African Journal of Food, Agriculture, Nutrition and Development*, vol. 17, no. 3, pp. 12198-12214.
- Adu-Gyamfi, J 2014, 'Childhood Construction and its Implications for Children's Participation in Ghana', *African Journal of Social Sciences*, vol. 4, no. 2, pp. 1-11.
- Agbenyegah, O & Gockwoski, J 2002, Labour Practices and the Cocoa Sector of Ghana with a Special Focus on the Role of Children. Findings from a 2001 Survey of Cocoa Producing Households., International Institute of Tropical Agriculture (IITA), Ibadan Nigeria.
- Ahmed, I 1999, 'Getting Rid of Child Labour', *Economic and Political Weekly*, vol. 34, no. 27, pp. 1815-1822.
- Akaba, S & Akuamoah-Boateng, S 2018, 'An Evaluation of Climate Change Effects on Fishermen and Adaption Strategies in Central Region, Ghana', in *Climate Change Impacts and Adaptation Strategies for Coastal Communities*, Springer, pp. 133-147.
- AMA 2008, Accra Metropolitan Assembly, Ablekuma South District Environmental Sanitation Strategy and Action Plan, Accra Metropolitan Assembly, Accra, Ghana, viewed 14th July, 2017, https://www.ircwash.org/resources/national-environmental-sanitation-strategy-and-action-plan-nessap-materials-transition-.
- Appleton, JV 1995, 'Analysing qualitative interview data: addressing issues of validity andreliability', *Journal of advanced nursing*, vol. 22, no. 5, pp. 993-997.
- Asuming-Brempong, S, Sarpong, D, Asenso-Okyere, K & Amoo, P 2007, Labour Practices in Cocoa Production in Ghana: National Programme for the Elimination of Worst Forms of Child Labour in Cocoa (NPECLC), Ministry of Manpower, Youth and Employment, Accra, viewed 14 July, 2017, <allafrica.com/stories/200811210849.html>.
- Bhalotra, S 2003, Child Labour in Africa, the OECD Social, Directorate for Employment, Labour and Social Affairs Employment, Labour and Social Affairs Committee, Employment and Migration Working Papers, 1815-199X, Paris, France, 28th April, 2003, pp. 1-78, <www.cgiar.org/,>.
- Bowlby, J 2008, A Secure Base: Parent-Child Attachment and Healthy Human Development, Basic books, New York, US.
- Bryman, A 2015, Social Research Methods, Oxford University Press, London, UK.
- Carter, N, Bryant-Lukosius, D, DiCenso, A, Blythe, J & Neville, AJ 2014, 'The use of triangulation in qualitative research', *Proceedings of the Oncology nursing forum*, vol. 41.
- Cofie, E 2015, 'Effects of Urbanization on Access to Livelihood Resources: Experiences of Farm Families in the Ga West Municipality, Ghana', University Of Ghana.
- Cooke, E, Hague, S & McKay, A 2016, *The Ghana Poverty and Inequality Report: Using the 6th Ghana Living Standards Survey*, University of Sussex, UNICEF, Ashesi University, Accra, Ghana, viewed 09th May, 2018, <www.africainequalities.org>.
- Cope, DG 2014, 'Methods and meanings: credibility and trustworthiness of qualitative research', *Proceedings of the Oncology nursing forum*, vol. 41.
- Denscombe, M 2014, *The good research guide: for small-scale social research projects*, McGraw-Hill Education (UK).
- DFID 1998, Ghana: Country Strategy Paper, Department for International Development (DFID), Department for International Development (DFID), DFID, 1 Palace Street, London SW1E 5HE, UK, viewed 15th July, 2017, <www.dfid.gov.uk>.

- Diallo, Y, Etienne, A & Mehran, F 2013, *Global Child Labour Trends 2008 to 2012*, International Labour Organization, International Labour Office, Geneva, Switzerland, viewed 07th May, 2018, http://www.ilo.org/ipec/Informationresources/WCMS_IPEC_PUB_23015/lang-en/index.htm.
- Draper, A & Swift, JA 2011, 'Qualitative research in nutrition and dietetics: data collection issues', *Journal of Human Nutrition and Dietetics*, vol. 24, no. 1, pp. 3-12.
- Ersado, L 2005, 'Child Labor and Schooling Decisions in Urban and Rural Areas: Comparative Evidence from Nepal, Peru, and Zimbabwe', *World Development*, vol. 33, no. 3, pp. 455-480.
- Figiel, S, Kuberska, D & Kufel, J 2014, *Agri-food Clusters in Poland*, Instytut Ekonomiki Rolnictwa i Gospodarki Żywnościowej-Państwowy Instytut Badawczy.
- Flora, C 2018, Rural Communities: Legacy+Change, Routledge, New York, US.
- Grainger, C & Parker, D 2013, 'The Political Economy of Fishery Reform', *Annual Review Resource Economics*, vol. 5, no. 1, pp. 369-386.
- Green, J & Thorogood, N 2018, *Qualitative Methods for Health Research*, Sage, 2455 Teller Road, Thousand Oaks, California 91320, US.
- GSS 2002, 2000 Population and Housing Census: Summary Report of Final Results, Ghana Statistical Service (GSS), Ghana Statistical Service (GSS), Accra, Ghana, <www.statsghana.gov.gh/docfiles/.../Census2010Summary_report_of_final_results.pd.>.
- --- 2007, *Patterns and Trends of Poverty in Ghana 1991-2006*, Ghana Statistical Service, Government of Ghana, Accra, Ghana.
- --- 2012, 2010 Population and Housing Census (PHC) Summary Report of Final Results, Ghana Statistical Service, Ghana Statistical Service, P.O. Box GP 1098 Accra, Ghana, May, 2012, pp. 1-103, <www.statsghana.gov.gh/>.
- --- 2014a, Ghana Living Standards Survey 6 with Labour Force Module (GLSS6/LFS) 2012/2013, Three-Cycle Labour Force Report, Ghana Statistical Service, Accra, Ghana, viewed January 03, 2017, <www.statsghana.gov.gh>.
- --- 2014b, Ghana Living Standards Survey Round 6 (GLSS 6), Child Labour Report, GSS, ILO, UNDP, UNICEF, DFID, World Bank, Accra, Ghana, viewed 06th May, 2018, <www.statsghana.gov.gh/docfiles/glss6/GLSS6_Child%20Labour%20Report.pdf>.
- --- 2015a, *Annual Gross Domestic Product: Ghana Statistical Service* GSS, Accra, Ghana, viewed January 14, 2018, <(http://www.statsghana.gov.gh/docfiles/GDP/GDP2015/2015_Annual_GDP_September _2015_Edition.pdf>.
- --- 2015b, *Ghana Poverty Mapping Report: Ghana Statstical Service*, Ghana Statistical Service, Accra, Ghana, pp. 1-77, viewed 19th August, 2017, <www.statsghana.gov.gh/>.
- Guarcello, L, Lyon, S & Valdivia, C 2015, Evolution of the Relationship between Child Labour and Education since 2000. Evidence from 19 Developing Countries. Understanding Children's Work Programme Working, Rome, Italy, viewed 12th May, 2018, http://unesdoc.unesco.org/images/0023/002324/232455e.pdf.
- Hashim, I 2007, 'Independent Child Migration and Education in Ghana', *Development and Change*, vol. 38, no. 5, pp. 911-931.
- 2005, 'Human Trafficking Act 694' Parliament of Ghana, Ghana.

- ICO 2016, 'The Global Market for Cocoa Products: Recent Trends and Future Outlook, International Cocoa Organization', paper presented at the ICCO Cocoa Market Outlook Conference, London, UK, September 27, 2016
- Minimum Age Convention, International Labour Organisation (ILO) 1973 (ILO) (Geneva, Swtzerland), viewed 21:09:2015, http://www.ilo.org/ilolex/cgilex/convde.pl?C138
- ILO, UNICEF & WB 2016, Understanding Children's Work: An Inter-Agency Research Cooperation Project ILO, UNICEF, and World Bank, Rome, Italy, July, 2016, pp. 1-38, viewed 22nd August, 2017, http://www.ucw-project.org/.
- Jim, O 1992, Community Psychology: Theory and Practice, Wiley Press, UK.
- Kabeer, N 1994, *Reversed Realities: Gender Hierarchies in Development Thought*, Verso, 6 Meard Street, London, W1F 0EG.
- Kaltenborn, M, Abdulai, A, Roelen, K & Hague, S 2017, *The Influence of Policy and Legal Frameworks on the Development of National Social Protection Systems*, vol. 2017 1781184062, no. 501, Institute of Development Studies (IDS), Centre for Social Protection (CSP), London, UK, viewed January 2, 2018, http://www.ids.ac.uk/idsresearch/centre-for-social-protection>.
- Kapoor, A 2017, Children at the Heart: Assessment of Child Labour and Child Slavery in Ghana's Cocoa Sector and Recommendations to Mondelez International, Mondelez International, Accra, Ghana, viewed 19th September, 2017, <www.embode.co>.
- Kral, M, Garcı'a, J, Aber, M, Masood, N, Dutta, U & Todd, N 2011, 'Culture and Community Psychology: Toward a Renewed and Reimagined Vision', *American Journal of Community Psychology*, vol. 47, pp. 46–57.
- Kuyini, A, Alhassan, A, Tollerud, I, Weld, H & Haruna, I 2009, 'Traditional Kinship Foster Care in Northern Ghana: The Experiences and Views of Children, Carers and Adults in Tamale', *Child & Family Social Work*, vol. 14, no. 4, pp. 440-449.
- Lim, J, Shon, E & Yang, E 2018, 'A qualitative study of healthcare providers' perceptions and suggestions for developing cancer survivorship care models in South Korea', *European journal of cancer care*, vol. 27, no. 1, p. e12613.
- Löwe, A 2017, Creating Opportunities for Young people in Ghana's cocoa Sector, The MasterCard Foundation, The Overseas Development Institute (ODI), Global Communities, Solidaridad, NCBA-CLUSA, and GOAL, Overseas Development Institute, 203 Blackfriars Road London SE1 8NJ, UK, viewed January 18, 2018, <www.odi.org>.
- Lucero, J, Wallerstein, N, Duran, B, Alegria, M, Greene-Moton, E, Israel, B, Kastelic, S, Magarati, M, Oetzel, J & Pearson, C 2018, 'Development of a Mixed Methods Investigation of Process and Outcomes of Community-based Participatory Research', *Journal of Mixed Methods Research*, vol. 12, no. 1, pp. 55-74.
- Marquette, C, Koranteng, K, Overå, R & Aryeetey, E 2002, 'Small-scale Fisheries, Population Dynamics, and Resource Use in Africa: The Case of Moree, Ghana', *AMBIO: A Journal of the Human Environment*, vol. 31, no. 4, pp. 324-336.
- Mason, J 2017, *Qualitative Researching*, Sage, 1 Oliver's Yard, 55 City Road, London EC1Y 1SP.
- Montgomery, H 2008, An Introduction to Childhood: Anthropological Perspectives on Children's Lives, John Wiley & Sons, Southern Gate, Chichester, West Sussex PO19 8SQ, UK.
- Mukherji, P & Albon, D 2018, Research Methods in Early Childhood: An Introductory Guide, 3 edn, Sage, 2455 Teller Road, Thousand Oaks, California 91320, US.

- Mull, L & Kirkhorn, S 2005, 'Child Labour in Ghana Cocoa Production: Focus upon Agricultural Tasks, Ergonomic Exposures, and Associated Injuries and Illnesses', *Public Health Reports*, vol. 120, no. 6, pp. 649–656.
- Nieuwenhuys, O 2005, Children's Lifeworlds: Gender, Welfare and Labour in the Developing World, Routledge Taylor and Francis, 29 West 35th Road, New York, NY 10001, US.
- Nolan, LB, Waldfogel, J & Wimer, C 2017, 'Long-Term Trends in Rural and Urban Poverty: New Insights Using a Historical Supplemental Poverty Measure', *The ANNALS of the American Academy of Political and Social Science*, vol. 672, no. 1, pp. 123-142.
- Ormston, R, Spencer, L, Barnard, M & Snape, D 2014, 'The foundations of qualitative research', *Qualitative research practice. A guide for social science students and researchers*, pp. 1-25.
- Pasanen, T 2016, A Snapshot of Out of School Youth, IDU, London, UK, viewed January 14, 2018, https://www.odi.org/opinion/10340-snapshot-out-school-youth-ghana).>.
- Post, D 2018, *Children's Work, Schooling, and Welfare in Latin America*, Routledge, 711 Third Avenue, New York, NY 10017, US.
- Quattri, M & Watkins, K 2016, *Child Labour and Education: A Survey of Slum Settlements in Dhaka*, Overseas Development Institute (ODI), Overseas Development Institute, 203 Blackfriars Road, London SE1 8NJ, UK, <www.odi.org>.
- Ray, R 2002, 'The Determinants of Child Labour and Child Schooling in Ghana', *Journal of African Economies*, vol. 11, no. 4, pp. 561-590.
- Richards, L 2014, Handling qualitative data: A practical guide, Sage.
- Ruckenstein, M 2014, 'Visualized and interacted life: Personal analytics and engagements with data doubles', *Societies*, vol. 4, no. 1, pp. 68-84.
- Ruf, F 2011, 'The Myth of Complex Cocoa Agroforests: The Case of Ghana', *Human Ecology* vol. 39, no. 3, pp. 373–388.
- SDG 2017, *United Nations Sustainable Development Goals Report* United Nations, New York, US, January 2, 2018, http://www.un.org/sustainabledevelopment/development-agenda/ >.
- Sen, A 1997, 'Human Capital and Human Capability', World Development, vol. 25.
- Sertich, M & Heemskerk, M 2011, 'Ghana's Human Trafficking Act: Successes and Shortcomings in Six Years of Implementation', *Human Rights Brief*, vol. 19, no. 1, pp. 1-6.
- Skoufias, E, Parker, S, Behrman, J & Pessino, C 2001, 'Conditional Cash Transfers and their Impact on Child Work and Schooling: Evidence from the Progresa Program in Mexico', *Economia*, vol. 2, no. 1, pp. 45-96.
- Sumberg, J, Anyidoho, N, Leavy, J, te Lintelo, D & Wellard, K 2012, 'Introduction: The Young People and Agriculture "Problem" in Africa', *IDS Bulletin*, vol. 43, no. 6, pp. 1–8.
- Trickett, E 2009, 'Community Psychology: Individuals and Interventions in community Context', *Annual Review of Psychology*, vol. 60, pp. 395–419.
- Tsikata, D 2010, 'Toward a Decent Work Regime for Informal Employment in Ghana: Some Preliminary Considerations', *Comparative Labor Law & Policy Journal*, vol. 32, pp. 311-342.
- Twum-Danso, A 2009, 'Reciprocity, Respect and Responsibility: The 3Rs Underlying Parent-Child Relationships in Ghana and the Implications for Children's Rights', *The International Journal of Children's Rights*, vol. 17, no. 3, pp. 415-432.

- USLD/TU 2015, Final Report of Survey Research on Child Labor in West African Cocoa Growing Areas, United States Department of Labor and School of Public Health and Tropical Medicine Tulane University, United States Department of Labor and School of Public Health and Tropical Medicine Tulane University.
- White, B 1999, 'Defining the Intolerable Child Work, Global Standards and Cultural Relativism', *Childhood*, vol. 6, no. 1, pp. 133-144.
- Ye, J 2017, 'Contours of Urban Diversity and Coexistence', *Geography Compass*, vol. 11, no. 9, pp. 1-8.
- Yirga, W 2012, 'Determinants of Access to Credit by Smallholder Farmers: The Case of Gozamen District, East Gojjam, Ethiopia', Haramaya University.
- Zelizer, V 2005, 'The Priceless Child Revisited', in *Studies in Modern Childhood*, Springer, Palgrave Macmillan, UK, pp. 184-200.

Highlights

- The study focuses on parental perceptions of the causes of child labour.
- The research is grounded on Concurrent Parallel mixed methods research design.
- Child labour has a multiplicity of causes including socialisation and poverty.
- This paper recommends the adoption of policies that would address WFCL.
- The link between economic deprivation and child labour must be considered.

